ORIGINAL RESEARCH

www.jvsgbi.com

The impact of the COronaVIrus Disease 2019 (COVID-19) pandemic on the clinical management of patients with vascular diseases: findings from Tier 3 of the COVID-19 Vascular sERvice (COVER) study

The Vascular and Endovascular Research Network COVER Study Collaborative, The Vascular and Endovascular Research Network (VERN)*

* Full author information can be found at the end of the article

Corresponding author:

Matthew Machin NIHR Academic Clinical Fellow in Vascular Surgery, Department of Surgery & Cancer, Imperial College London, London, W6 8RF, UK Email: matthew.machin12@ imperial.ac.uk.

Received: 29th July 2022 Accepted: 4th August 2022 Online: 27th October 2022

Plain English Summary

Why we undertook the work: COVID-19 caused major disruption to the ability of hospitals to provide normal care for patients suffering from vascular conditions. The aim of this study was to evaluate how the pandemic affected the treatment delivered to patients with vascular conditions. It took place over a 4-week period during the first wave of the pandemic.

What we did: Vascular surgeons in 19 countries took part, collecting information from 1,801 patients. For every patient referred urgently to the emergency on-call vascular team during the period, the treatment provided was recorded and if/how that differed from the ideal treatment plan. Treatment differences were graded from 1 to 5 to illustrate how serious the changes were. Minor changes, such as a different imaging technique used, scored 1. Major life-changing differences, such as amputation where otherwise a patient would have been treated with an operation to save their limb, were scored as 5.

What we found: The study found that globally as many as one in five people received a different treatment compared with what they would have had prior to the pandemic. This included one in three people with abdominal aortic aneurysm, one in five people with carotid stenosis and one in 10 people with diabetic foot disease. One-third of changes to care were classed as major, the most common being non-surgical treatment or delayed surgery for a patient who would normally have received immediate or urgent surgery. Very rarely, clinical care was improved such as the increased access to urgent hot clinics, resulting in shorter waiting time and admission avoidance, for those with diabetic foot infections.

What this means: This study demonstrates that many patients did not receive ideal care for key vascular conditions during the first pandemic wave. We can see that the policy changes and updates to guidelines were rapidly implemented and that hospitals were responsive to these changes. Interestingly, as many as one in five of the changes from ideal care were related to patients' avoidance of hospitals for fear of catching coronavirus. If coronavirus-free surgical hubs are to be used going forward, patient perception of the safety of these centres will be key to their success.

Abstract

Background: During the first wave of coronavirus disease-19 (COVID-19), the pressure on healthcare led to significant restrictions and rationing of available services. Globally, vascular services were forced to change clinical management for a range of common, acute vascular presentations. The COVID-19 Vascular sERvice (COVER) Study Tier 3 aimed to understand the severity and impact of those changes on vascular patient referrals and acute presentations as a result of the pandemic in a global vascular cohort.

Methods: A 4-week multicentre multinational prospective observational study was launched in March 2020. Any hospital accepting urgent on-call vascular referrals during the pandemic was eligible to participate. Clinicians were asked to outline their actual management plan for each individual patient and to identify if the COVID-19 pandemic had changed their decision making. If so, clinicians then outlined their 'ideal' (pre-pandemic) management plan. All cases where management plans differed from that of the 'ideal' pre-pandemic management underwent further analysis to quantify the severity of the alteration on a scale of 1–5 (1: minor deviation in management, including imaging modality change, 5: palliation or amputation, where otherwise a patient would have been treated).

Results: A total of 1,801 patient episodes from 52 centres in 19 countries were included. The most commonly referred vascular conditions were chronic limb-threatening ischaemia (n=517, 28.7%), diabetic foot infections (n=237, 13.2%) and acute limb ischaemia (n=224, 12.4%). Only 2.3% of patients had a confirmed diagnosis of COVID-19. Deviations in management from pre-pandemic treatment occurred in 34.8% of those with unruptured \geq 5.5 cm diameter abdominal aortic aneurysms, 27.0% of people with symptomatic carotid artery disease, 17% of people with acute or chronic limb-threatening ischaemia and 12.7% of people with diabetic foot conditions. Of these modifications, 40.7% were categorised as significant (grade 3a/3b) and 38.1% as major (grade 4), such as non-operative instead of operative management of carotid and lower limb disease. Life-changing/ending plans including major amputation or palliation, where pre-pandemic patients would have been offered limb or life salvage procedures, were made in 4.9% of cases (grade 5).

Lessons learnt: The results of this study suggest that the clinical disruption experienced by vascular surgery patients during the COVID-19 pandemic was largely due to redistribution of resources rather than individual patient infection. Guidelines for service modifications were adhered to, leading to significant changes in clinical management. One in five episodes of change in management were due to patient avoidance of clinical areas for fear of contracting coronavirus. If surgical hubs, with a low risk of COVID-19 infection, are to be utilised, then patient perception of the safety of these centres will be key to their success.

Conclusion: The COVER Study Tier 3 was able to show that close to one in five clinical vascular management plans were affected globally during the pandemic, with a small proportion leading to life-changing outcomes.

Key words: vascular surgery, cardiovascular, COVID-19, coronavirus

Introduction

The COronaVIrus 2019 (COVID-19) pandemic resulted in major changes in medical practice and clinical decision-making. Throughout the pandemic, the decision of whether, and how, to offer medical or surgical treatments was influenced by several factors such as COVID-19 positivity, patients' risk of contracting COVID-19 whilst receiving treatment, resource rationing, staff shortages and government guidelines.^{1,2}

These considerations were pertinent for vascular patients as they typically have multiple comorbidities. Furthermore, vascular procedures are usually offered in tertiary centres, which were also hubs for managing COVID-19 patients, and suffered significant pressures on available resources.

To support vascular surgeons with decision making in resourcescarce settings, the international vascular community produced guidance for clinicians on how to manage patients during the height of the pandemic. One example was the Vascular Society of Great Britain and Ireland (VSGBI) COVID-19 guidance, produced in early 2020, recommending groups for delayed or conservative intervention during periods of greatest resource scarcity (presentations who would have received prompt and guidelinerecommended surgery prior to the pandemic).³

The impact of COVID-19 across countries or regions differed greatly in terms of both the timing and the number of people affected. As a result, the scale of the changes to vascular patients' treatment and decision-making during the COVID-19 pandemic remains unknown. The COVID-19 Vascular sERvice (COVER) Study was a prospective multinational observational study

evaluating the impact of COVID-19 on vascular patients using qualitative and quantitative methodology across 53 countries.⁴ This report explores the findings from Tier 3 of the COVER Study. The aim was to examine decision-making for new referrals made to vascular units during the first wave of the pandemic and compare the care given with 'ideal' (or pre-pandemic) practice.

Methods

Study design

The COVER study was an international prospective cohort study (ISRCTN registration reference number: 80453162) designed and coordinated by the Vascular and Endovascular Research Network (VERN). The study protocol is available online and was published prior to commencing recruitment.⁴ Tier 3 of the COVER study presented here evaluated the type of care provided to patients who presented with a vascular pathology during the first wave of the pandemic. All referrals made to the emergency on-call vascular service were eligible patient episodes, even if the referred condition was subsequently deemed non-urgent. Any surgeon providing direct vascular care in a hospital setting was eligible to participate. Invitations were sent via email and social media in March 2020 by the VERN group. A total of 52 centres across 19 countries took part in the study.

Approvals

Ethical approval for the COVER Study was provided by a UK National Health Service (NHS) Research Ethics Committee (REC) and the UK Health Research Authority (HRA) (reference: 20/NW/0196 Liverpool Central). All regulatory approvals were granted prior to commencing patient recruitment in late March 2020. Non-UK centres obtained institutional review board approval (locally or regionally) according to their individual policies and in compliance with the Declaration of Helsinki.

Participants

Any patient with an emergency or urgent presentation referred to the participating institutions' vascular services was eligible for inclusion, regardless of age or mode of referral. Participation in the study did not alter patient care in any way.

Data collection

Detailed records of how surgeons' practice deviated from prepandemic care were completed on a per-patient basis. Electronic remote data collection proformas were designed, tested and refined by the VERN COVER team in March 2020. Baseline data for each patient were captured prospectively upon patient referral or presentation. This included demographics, type of condition or nature of referral, co-existing health conditions, medications, American Society of Anesthesiologists (ASA) physical status classification, frailty, as well as details relating to the presenting vascular complaint.

Clinicians were asked to report their current plan using a multiple options list specifically designed for each of the vascular conditions. This included information on whether treatment occurred as an inpatient or outpatient, whether patients underwent non-invasive or invasive imaging, and the nature of any endovascular or operative intervention(s). Clinicians were then asked whether their plan for each patient had been modified or impacted upon due to constraints specifically related to the pandemic. If so, they were also asked to detail their 'ideal' management plan as it would have been pre-pandemic, using the same list of multiple-choice options.

A vascular change in practice specific score was developed by the study team to permit reporting of the severity of the change in practice (see Supplementary figures at www.jvsgbi.com) for all cases with a reported change of management plan.

The VERN COVER team (12 vascular specialists from centres, UK-based) reviewed all possible iterations of management changes for each clinical condition and scored each change using the following grading system:

- Minor deviation in management eg, imaging modality change (provided recognised form of imaging); changing which oral medication is prescribed; changing the intravenous medication prescribed.
- Minor deviation in management, which can have a more significant impact – eg, changing of medication from intravenous to oral; change in outpatient movement/flow, including seen in a different clinic; not being discussed in MDT; not having assessment of fitness (pulmonary function testing); not referring to specialist when otherwise would have.

- Deviation from practice eg, being managed as an outpatient when would have normally been inpatient; increase in threshold of intervention (for abdominal aortic aneurysm (AAA) surgery).
 - Less significant deviation in surgical/interventional practice or threshold – eg, ward-based amputation versus theatrebased amputation; slight technical difference in treatment (ie, different type of surgical bypass, different type of endovascular intervention).
 - b. More significant deviation eg, endovascular intervention rather than open surgery for treatment of AAA.
- Major deviation from practice eg, amputation versus revascularisation; no revascularisation versus revascularisation; major amputation versus minor amputation.
- 5. Palliation or major amputation, where otherwise a patient would have been treated.

If there were multiple changes to a patient's management pathway, they were allocated a score based on the single most significant and severe change.

Data management

De-identified data were transferred to a secure UK NHS server based at the University of Birmingham. Data sharing agreements with each participating centre were in place to ensure data protection. Each centre was required to record local identifiers on a secure General Data Protection Regulation (GDPR)-compliant database to allow longitudinal data capture and linkage, overseen by the Research and Development Department of the University Hospitals of Coventry and Warwickshire, Coventry, UK, who also acted as study sponsor. No identifiable patient data were recorded or exchanged.

Data handling and analysis

Each participating centre identified and entered consecutive patient data over a single 4-week period; the start of the four-week time period was selected at the discretion of each centre after all necessary approvals were in place. A data lock was applied on 9th September 2020. Data are reported as raw frequencies and percentages. Normally distributed data have been presented as mean ± standard deviation (SD) and non-normally distributed data are presented as median (range) values. Statistical analysis was performed using SPSS version 26.0 (SPSS, IBM, Chicago, Illinois, USA) and the language and statistical computing software R (V6) utilising additional software libraries. Independent variables were treated as nominal factors. Numerical data were treated as continuous data to maintain power. Primary and secondary outcomes were treated as dichotomous variables. Completeness of data was interrogated with a threshold of 90% for participants and 70% for dependent variables. Logistic regression was undertaken for the primary outcome. Univariate analysis was performed on all independent variables as a prerequisite for inclusion in the multivariate analysis, with a set threshold of p<0.1.



Results

A total of 1,801 patient episodes were captured from 52 centres across 19 countries over a 4-week period (Figure 1). The largest proportions of referrals were made by speciality services such as secondary care physicians and podiatrists (Figure 2). Referrals from an emergency department constituted 26.6% of referrals. The most common condition referred to vascular clinicians was chronic limb-threatening ischaemia (28.8%; Table 1).

A total of 41 patient episodes had a confirmed diagnosis of COVID-19 infection, 2.3% of the total 1801 episodes. Most patients were male (69%) and had diagnoses of hypertension and/or diabetes mellitus (Table 2). A pre-existing diagnosis of peripheral arterial disease was present in 31.5% of participants. Previous myocardial infarction was documented in 18.2% of patients, congestive cardiac failure in 7.8%, and/or angina in 5.5%.

Table 3 shows a summary of the changes in management for each of the presenting vascular conditions. We present the specific findings for 1334 patients presenting with the following condition areas: AAA, acute aortic syndromes, acute limb ischaemia, carotid artery disease, chronic limb-threatening ischaemia and diabetic foot infection or ulceration (see Appendix 1 at www.jvsgbi.com for details of presentations across conditions).

Deviation from 'ideal' management

A total of 290 patient episodes (16.1%) had a confirmed change from the 'ideal' management plan. The highest proportion of patient



Figure 2 Mode of referral to specialist vascular surgery services for the included participants.

Table 1 The proportions of conditions referred to	vascular
specialists during the study ranking.	

Condition referred	n (% of 1,801)
AAA <5.5 cm in maximal diameter, asymptomatic or non-urgent	31 (1.7%)
AAA ≥5.5 cm in maximal diameter, asymptomatic or non-urgent	66 (3.7%)
Aortic aneurysm urgent (eg, ruptured or patient presenting with pain)	50 (2.8%)
TBAD or thoracoabdominal aneurysm or acute aortic syndrome or traumatic aortic injury	47 (2.6%)
Severe claudication or CLTI	517 (28.7%)
Acute limb ischaemia	224 (12.4%)
Asymptomatic carotid artery disease (atherosclerotic)	33 (1.8%)
Symptomatic carotid artery disease (atherosclerotic)	129 (7.2%)
Diabetic foot infection or ulceration	237 (13.2%)
lliofemoral deep vein thrombosis	37 (2.1%)
Venous leg ulceration	56 (3.1%)
Other	374 (20.8%)

AAA, abdominal aortic aneurysm; TBAD, type B aortic dissection; CLTI, chronic limb-threatening ischaemia.

Table 2 Baseline	characteristics	of the	included	participants.	

Characteristic	n (% of 1,801)
Male gender	1,234 (68.5%)
Peripheral arterial disease	568 (31.5%)
Diabetes mellitus	753 (41.8%)
Hypertension	1,018 (56.5%)
Chronic obstructive pulmonary disease	283 (15.7%)
Lower respiratory tract infection, including suspected COVID-19 (but not laboratory confirmed)	22 (1.2%)
Confirmed COVID-19	41 (2.3%)
Myocardial infarction	327 (18.2%)
Angina	99 (5.5%)
Heart failure	141 (7.8%)
Atrial fibrillation or dysrhythmia	287 (15.9%)
Stroke or transient ischaemic attack	261 (14.5%)
End stage renal disease on dialysis	116 (6.4%)
Chronic kidney disease	252 (14.0%)
Actively smokes tobacco cigarettes	330 (18.3%)
Cancer	121 (6.7%)
Dementia	72 (4.0%)
COVID-19, COronaVIrus Disease 2019.	

episodes in which a change in management occurred was for those with AAA >5.5 cm in size (34.8%).

Regarding the severity of change in management, the proportion of patient episodes graded as 1, 2, 3a, 3b, 4 and 5 were 4.9%, 10.3%, 23.8%, 17.5%, 38.6% and 4.9%, respectively.

Aortic referrals

Overall, 30 patient episodes out of a total of 196 (15.3%) for those with aortic pathologies underwent a change in management. There were no changes in the management of the 31 sub-treatment threshold (<5.5 cm) AAA referrals. Of the 66 patients with an AAA measuring >5.5 cm, 34.8% (n=23) had a significant deviation in management; 14 (21.2%) scored 3, including a change from open to endovascular AAA repair (EVAR) despite suboptimal anatomy or a lack of cardiopulmonary exercise testing and anaesthetic review prior to treatment. Nine patients (13.6%) had a major deviation in management (score 4) which was predominantly an increase in the threshold to offer surgical treatment or delay to repair indefinitely. There were 50 ruptured or symptomatic AAA referrals, of which four (8%) had a major deviation (score 4 or 5) to their care compared with standard pre-pandemic practice: three (6%) underwent emergency open repair where EVAR would have been preferred but suitable theatre space or availability was limited, and in one case (2%) there was an issue with the endograft supply chain.

Forty-seven cases of acute aortic syndrome were referred but only three patients (6.4%) experienced a change to their planned management in the form of postponement of thoracic EVAR.

Acute limb ischaemia

A change in management occurred in 17% of acute limb ischaemia referrals (38 of 224 patients). Five patients (2.2%) with unsalvageable limbs received no intervention where, ideally, they would have undergone a major lower limb amputation (score 4). Six patients (2.7%) had a 'major' (score 4) change to their management which no longer involved an operative or interventional procedure (where 'ideally' indicated). The remaining changes were in the most part an alternative intervention or imaging modality, or a delay to intervention (score 2 and 3a).

Carotid artery disease

Overall, 40 patient episodes out of a total of 162 (24.7%) for those with carotid artery pathologies underwent a change in management. Major (level 4) changes to management were documented in 15.2% of patients, who were not offered a carotid endarterectomy (CEA) when they would have been pre-pandemic. There were 129 symptomatic carotid disease referrals with 35 (27%) having a documented change in management. Nineteen (14.7%) had a major change (score 4) and were offered best medical therapy only rather than either CEA or carotid artery stenting. Significant modifications (scores 3a and 3b) included delays to intervention beyond the recommended 14 days.

Condition	Total n	Number of management plans changed	Severity of change to clinical management		
			Score/Class	N (%)	
		0.(01/)			
AAA <5.5 cm	31	0 (0%)	N/A	N/A	
VAA >5.5 cm	66	23 (34.8%)	3b	14 (21.2%)	
			4	9 (13.6%)	
uptured/symptomatic AAA	50	4 (8.0%)	4	4 (8%)	
cute aortic syndrome	47	3 (6.4%)	3b	2 (4.2%)	
			4	1 (2.1%)	
cute limb ischaemia	224	38 (17.0%)	1	2 (0.9%)	
			2	6 (2.7%)	
			3a	13 (5.8%)	
			3b	4 (1.8%)	
			4	6 (2.7%)	
			5	5 (2.2%)	
			4 (positive)	2 (0.9%)	
symptomatic carotid presentations	33	5 (15.2%)	4	5 (15.2%)	
Symptomatic carotid presentations	129	35 (27.0%)	1	1 (0.8%)	
			2	3 (2.3%)	
			3a	9 (7.0%)	
			3b	3 (2.3%)	
			4	19 (14.7%)	
LTI	517	88 (17.0%)	1	8 (1.6%)	
			2	11 (2.1%)	
			3	9 (1.7%)	
			За	15 (2.9%)	
			3b	14 (2.7%)	
			4	30 (5.8%)	
			5	1 (0.2%)	
iabetic foot infection	237	30 (12.7%)	2	3 (1.3%)	
			За	7 (3.0%)	
			3b	2 (0.84%)	
			4	12 (5.1%)	
			5	5 (2.11%)	
				\ /	

Table 3 Alterations to management of the five key conditions (n=1,334)

AAA, abdominal aortic aneurysm; CLTI, chronic limb-threatening ischaemia.

1. Minor deviation in management – eg, imaging modality change (provided recognised form of imaging); changing which oral medication is prescribed; changing the intravenous medication prescribed.

- 2. Minor deviation in management, which can have a more significant impact eg, changing of medication from intravenous to oral; change in outpatient movement/flow, including seen in a different clinic; not being discussed in MDT; not having assessment of fitness (pulmonary function testing); not referring to specialist when otherwise would have.
- Deviation from practice eg, being managed as an outpatient when would have normally been inpatient; increase in threshold of intervention (for abdominal aortic aneurysm (AAA) surgery).
 a. Less significant deviation in surgical/interventional practice or threshold eg, ward-based amputation versus theatre-based amputation; slight technical difference in treatment (ie, different
 - type of surgical bypass, different type of endovascular intervention).
 - b. More significant deviation eg, endovascular intervention rather than open surgery for treatment of AAA.
- 4. Major deviation from practice eg, amputation versus revascularisation; no revascularisation versus revascularisation; major amputation versus minor amputation.
- 5. Palliation or major amputation, where otherwise a patient would have been treated.

Chronic limb-threatening ischaemia

Of the 517 presentations, 88 (17.0%) received an alternative management plan to documented pre-pandemic ideal care. Half of these (45 of 88) had a significant or major adjustment (score 3a, 3b or 4), with one being palliated rather than being offered intervention (score 5). These significant or major deviations included delays to treatment and changing the operative modality to endovascular rather than open due to unavailability of endovascular suites or vice versa due to a lack of theatre space.

Diabetic foot infection or ulceration

Of the 237 patients seen with a diabetic foot infection or ulceration, 30 (12.7%) had a change in management. This included rare positive changes to care, such as access to urgent clinic assessment facilitated by a reduction in elective or other work, described in a single unit (positive 3a). For the remaining cases, the changes were adverse. Five cases (2.1%) had a major lower limb amputation (score 5) due to delays in presentation and access to services, which may have been avoidable if treated earlier. Twelve patients (5.1%) had wound and foot care only when 'ideally' revascularisation would have been indicated, constituting a 'major' modification (score 4). Ten patients underwent a 'significant' alteration (score 3a or 3b) to their plan with only ward-based foot care, toe amputation or sepsis drainage instead of operating theatre-based care under general or regional anaesthesia.

Drivers of change to management

The clinicians' reasons for having to provide management different from their ideal care were varied. A total of 240 patient episodes were confirmed to have a change in management across the chronic limb-threatening ischaemia, acute limb ischaemia, diabetic foot infection, carotid artery disease, AAA and acute aortic syndromes.

The barriers and stimuli for changes in management for these (grouped) conditions are listed in Table 4. The most common was department policy change for specific conditions (55.4%). The second most common reason was selected as "Other", with 46.3% of patient episodes listing this. 18.8% of patient episodes with confirmed altered management listed "Patient declined due to risk of hospital-acquired COVID-19". Other factors included a lack of operating space, lack of anaesthetic availability and a lack of suitable high-level care postoperative beds. Confirmed COVID-19 infection as a reason to change management was rare but did influence management negatively where indicated.

Discussion

Emerging single institution and surgeon surveys published during the first and subsequent waves of infection demonstrated significant variation in how vascular services managed the interruption to normal services, such as how surgery was rationed and the move from inpatient to outpatient management to reduce pressure on hospital beds.^{5,6} The COVER Tier 3 study has

 Table 4 Barriers to ideal management encountered for all patient

 episodes in which there was a change in management.

Reasons listed as barrier to ideal management	n (% of 240)
Departmental policy	133 (55.4%
Lack of imaging capacity	24 (10.0%)
Patient declined due to risk of hospital-acquired COVID-19	45 (18.8%)
Patient is COVID-19 positive	18 (7.5%)
No vascular beds for patient to be admitted to	17 (7.1%)
No postoperative bed for patient to return to	11 (4.6%)
Insufficient staff (eg, illness, redeployment)	24 (10.0%)
No interventional radiology procedural capacity	22 (9.2%)
No anaesthetic pre-assessment availability/cardiopulmonary testing	12 (5.0%)
No level 2/3 hospital bed (admitting capacity)	19 (7.9%)
Other	111 (46.3%)
240 is the total number of patient episodes in which management was c	onfirmed to have

240 is the total number of patient episodes in which management was confirmed to have changed due to COVID-19 pandemic in the following conditions: AAA <5.5 cm, AAA >5.5 cm, ruptured AAA, acute aortic syndromes, acute limb ischaemia, chronic limb-threatening ischaemia, diabetic foot infection.

AAA, abdominal aortic aneurysm.

described these changes in more detail, and across a variety of countries and vascular centres.

As expected, lower limb emergencies comprised the most commonly referred condition, making up over half of the cases in this study (chronic limb-threatening ischaemia, acute limb ischaemia and diabetic foot infection/ulceration). The low proportion of emergency aortic pathology may reflect the global downward trend in AAA presentations, but does not take into account patients who died before reaching hospital or those who died at home while isolating.⁷ A change in referring practices from stroke teams in response to guidelines recommending medical management for all symptomatic carotids except for crescendo transient ischaemic attacks may have led to a lower proportion of those patients being picked up by this study.³

Uniquely, Tier 3 of the COVER study has attempted to quantify the effect of the COVID-19 pandemic on the management of individual vascular referrals. Data indicated that around one in three above threshold AAA and symptomatic carotid disease patients presenting during the pandemic were delayed. With AAA rupture rates increasing incrementally beyond 5.5 cm, and between one in six to one in 13 strokes thought to be prevented by carotid intervention for high-grade symptomatic stenosis, the potential loss of quality-adjusted life years could be high in the coming years.⁸ One in 10 chronic limb-threatening ischaemia presentations suffered delays and changes to management, and diabetic foot patients also suffered from delays to revascularisation and debridement. Guidelines have shown that non-optimised care in these groups leads to higher rates of major amputation and rising healthcare costs.⁹ This will place an additional burden on healthcare services globally.

The most common reason identified for a change in management was departmental policy, followed by "Other". Preemptive action was taken in many countries to suspend elective activities and redeploy specialty surgical staff to intensive care and respiratory services prior to the peak incidence of COVID-19, reflecting the stimuli for change being departmental policy rather than individual patient-related or clinical factors. Selection of "Other" likely reflects the complex decision making, with multiple factors contributing to a change in management that cannot be put down as a restriction or lack of any given resource.

Patient perception of hospital-acquired COVID-19 was a barrier in as many as one in five patient episodes. If COVID-free surgical hubs are to be utilised effectively in the future, it will be important to address perception to ensure effective delivery of elective services.

As reflected in the prevalence of COVID-19 in this cohort, individual infection with COVID-19 was a barrier in as few as 7.5% of cases. This again reiterates that the vast majority of changes in management of vascular patients stemmed from a shift in service provision.

Limitations of the study

Whilst this study was designed with the intention of capturing all referrals to acute vascular services and to document the management plans accordingly, it is likely that there will have been a proportion of missed urgent cases, especially those in the community who never made it to the point of referral. The busy vascular take, coupled with workforce pressures and a shift to tele- or remote services, 1,10 may mean that referrals received electronically or those triaged directly to outpatient clinics might not have been documented. Urgent or emergency vascular conditions, in general, carry a high associated mortality, in particular ruptured AAA or complicated aortic dissection. It is therefore possible that some cases may have been missed due to a lack of referral resulting from delays in presenting to medical services due to isolation or fear of catching COVID-19 in hospital. The timing for the 4-week data capture window was pragmatic and left to the participating institution to decide the start date. Consequently, in some cases this may not have aligned with the 'true' peak of the COVID-19 pandemic and could lead to under-representation of the true impact of clinical management changes. Long-term data out to 1 year from presentation was insufficient to provide information on impact on longer-term outcomes.

Furthermore, the scoring system used to grade the severity of change has not been validated. It was developed by UK-based vascular specialists and therefore may be biased towards UK-based practice.

Lessons learnt

The results of this study suggest that the clinical disruption

KEY MESSAGES

- During the COVID-19 pandemic, one in five vascular clinical management plans were affected globally.
- Only a small proportion of patients had an immediate life or limb changing outcome (major amputation or palliation).
- Many of these alterations in management involved deferrals of treatment and will influence waiting lists for the foreseeable future.

experienced by vascular surgery patients during the COVID-19 pandemic was largely due to redistribution of resources rather than individual patient infection. The guidelines for service modifications, which were rapidly produced and published by the Vascular Society of Great Britain and Ireland, appear to have been taken up and adhered to both within the UK and further afield, leading to very real changes in clinical management during the first pandemic wave. Refinement of these guidelines should be a priority going forward, so that plans for service modification in the event of a future pandemic are available to expedite their timely publication.

It was interesting to note that one in five episodes of change in management were due to patient avoidance of clinical areas for fear of contracting coronavirus. If surgical hubs, with a low risk of COVID-19 infection, are to be used, then patient perception of the safety of these centres will therefore be key to their success.

Unfortunately, the lack of outcome data means it is difficult to interpret the impact of these changes in practice, and this is a limitation of the study.

Conclusion

Tier 3 of the COVER Study is unique in its description of management changes and the granularity of those changes. We have shown that decisions made for nearly one in five patients presenting during the pandemic were affected by significant or lifechanging/ending alteration(s). Clearly these changes across the breadth of vascular surgery will lead to a surge of patients whose management was temporised in the initial pandemic wave.

Conflict of Interest: None.

Funding: The funding sources and the study sponsor had no role in study design, collection, analysis or interpretation of the data.

Acknowledgements: VERN would like to acknowledge the support of Sonia Kandola and the University Hospitals of Coventry and Warwickshire Research and Development Department for study sponsorship and oversight. In addition, we would also like to thank our funders the Circulation Foundation and the international collaborating organisations.

Authorship:

Writing committee: Sandip Nandhra, Ruth A Benson, Matthew Machin, Rachel Forsythe, Joseph Shalhoub and Athanasios Saratzis. Critical manuscript review: Brenig Gwilym, Rachel Forsythe, Nikesh Dattani, Graeme K Ambler, Ryan Preece, George Dovell, Sarah Onida, Aminder Singh, Panagiota Birmpili, Rob Blair, David C Bosanquet, Sarah Aitken, Andrew Choong, Adam Johnson, Jana-Lee Moss, Jun Jie Ng and Chris Imray

Study Steering committee:Ruth A Benson and Sandip Nandhra (study co-leads). Joseph Shalhoub, Graeme K Ambler, Nikesh Dattani, David C Bosanquet, Rachael Forsythe, Sarah Onida, George Dovell, Louise Hitchman, Ryan Preece, Athanasios Saratzis (co-chief investigator), Chris Imray (co-chief investigator)

International steering team: United States of America: Adam Johnson

Asia (Hong Kong SAR China/Malaysia/Singapore): Andrew Choong, Jun Jie Ng Australia and New Zealand: Sarah Aitken, Jana-Lee Moss

Data analysis: Sandip Nandhra, Ruth Benson, Matthew Machin, Graeme K Ambler, Dave Bosanquet

Tier 3 database management and quality assurance:Ruth A Benson, Sandip Nandhra, Brett Dawson

Study communications committee: Rachael Forsythe, Ryan Preece, Louise Hitchman

Collaborator Authorship: Amr Abdelhaliem, Aseel Abuduruk, Thomas M Aherne, Hazem Ahmed, Tasleem Akhtar, Bekir Bogachan Akkaya, Julien Al Shakarchi, Abeer Waleed Alfardan, Abdeljawad J Algasi, Musaad AlHamzah, Ahmed Abdulrahman Alhumiad, Wissam Al-Jundi, Bernard Allard, Meshal Almeshal, Meshal Doham Almeshal, Faris Alomran, Reem Nasser AlRakaf, Mohamed Altabal, Abdulmajeed Hamad Altoijry, Talal Altuwaijri, Nasser Alwehaibi, Domenico Angiletta, Afroditi Antoniou, George A Antoniou, Libnah Leal Areias, James Ashcroft, Doaa Attia, Lukas Attwell, Mohammed A Azab, Omar Aziz, Ahmed Y Azzam, Efstratia Baili, Danielle R Bajakian, Sara Joy Anderson Baker, Christos Bakoyiannis, Hashem Barakat, Khalid Bashar, Ruth Battersby, Shanka Benaragama, Ahmed Tarek Saeed BenGhatnsh, Nikolaos Bessias, Resya Bhakthavalsalan, Shagran Binkhamis, Roshan Bootun, Emily Boyle, Ion Buga, Robert E Carlin, Martin Catterson, Jennifer Chambers, Emily Chan, Karishma Chandarana, Alexandros Charalabopoulos, Gabriella Charlton, Ankur Chawla, Stephen Wing Keung Cheng, Natasha Chinai, Asad J Choudhry, Annie Clothier, Tina U Cohnert, Chloe Coleman, Rebeca Magabeira Correia, Michael Costanza, Patrick Anthony Coughlin, James Coulston, James Cragg, Katy Darvall, Emma Mary Davies, Huw Davies, Meryl Davis, Claire Dawkins, Joseph Dawson, Anastasia Dean, José Maciel Caldas dos Reis, Andrew Duncan, Mark Edwards, Bridget Egan, Mehdi El Amrani, Mohamed Gamal Elbahnasawy, Muhammed Elhadi, Ahmed Elhadi, Mohamed Saleh Eljareh, Ramy Elkady, Mohamed Elkawafi, Fatimah S Elkhafeefi, Maysoon Elkwahad, Ibrahim Abdelghafar Ellojli, Kareem ElSanhoury, Hazem Elsantawy, Khaled Elsayed, Mohamed D Elsheikh, Raed M Ennab, Anthony C Feghali, Owain Fisher, Ronald Luiz Gomes Flumignan, Amy L Fowler, Richard Francis Galloway, Andrew Garnham, Sotirios Georgopoulos, Tamer MH Ghatwary Tantawy, Ravi Raj Goel, Mingzheng Aaron Goh, Tabitha Grainger, Nalaka Gunawansa, Farag Saad Mohamed Hafed, Eric E Hammond, Simon Christopher Hardy, Thomas James Hardy, Gareth James Harrison, Ahmed Hassanin, Hytham KS Hmaid, Bella Huasen, Madeleine Dulany Hunter, Sabir Ibrahim, Chris Imray, Kaisor Iqbal, Hakkı Zafer Iscan, Arda Isik, Adam P Johnson, Doireann Patricia Joyce, Maciej Juszczak, Kiriaki Kakavia, Stavros K Kakkos, Christos D Karkos, Emmanuel Katsogridakis, Rana Khalil, Andrew Irwin Khallaf, Aazeb Khan, Manar Khashram, Samantha Khoo, Beatrice Kuang, Ganesh Kuhan, Ioanna Kyrou, Pierfrancesco Lapolla, Eunice Lim, Dafydd Locker, Xun Luo, Oliver Timothy, Auban Lyons, Ragai R Makar, Dimitris Maras, Martin Maresch, Emmeline A Martin, Hayrettin Levent Mavioglu, Dennis Mazingi, James McCaslin, David McClure, Kevin McKevitt, Lewis Meecham, Shreya Mehta, Fabrizio Minelli, Andrea Mingoli, Afroditi ME Mitka, Farag S Mohamed, Hayley Marie Moore, Rachael Lucia Morley, Jana-Lee Moss, Konstantinos G Moulakakis, Ahmed Msherghi, Kamel Muhammad, Luis Carlos Uta Nakano, Mohamed Ezzat Nassreldin, Craig Nesbitt, Jonathan Nicholls, Andrew Nickinson, Thamer Nouh, Jose MS Nunag, Isaac K Nyamekye, Isam Osman, Theofanis T Papas, Konstantinos O Papazoglou, Sharath Paravastu, Noala Parr, Ketino Pasenidou, Franklin Pond, Matthew Adam Popplewell, Katarzyna Powezka, Daniela Price, Sivaram Premnath, Raffaele Pulli, Hussein MM Rabee, Habibur P Rahman, Nandhini Ravintharan, Nicole C Rich, Toby Richards, Konstantinos Roditis, Alexander Eric Slavko Rolls, Iain Nicholas Roy, Hani Saeed, Prakash Saha, Alberto Saltiel, Louise Sanderson, Brena Costa Santos, Paolo Sapienza, Emma Scott, Christopher Selvaraj, Atif Sharif, Palma Shaw, Lauren

Shelmerdine, Simona Sica, Justinas Silickas, Sarah Silito, Aminder Anthony Singh, Gurkirat Singh, Ashwin Sivaharan, Pranav Somaiya, Gerry Stansby, Abhilash Sudarsanam, Scott M Surowiec, Elizabeth Suthers, Helen Suttenwood, Ahmed Taha, Mohamed Abdalla Hifny Taha, Siu Chung Tam, Alethea Meiyan Tang, Robert Tang, Dana Taran, Lawrence Tarusan, Myat Soe Thet, Jacqueline Thomas, Sean Tierney, Konstantinos Tigkiropoulos, Giovanni Tinelli, Mahmoud MH Tolba, Hannah C Travers, Ioannis Tsagkos, Yamume Tshomba, Paraskevi Tsiantoula, Christopher Paul Twine, Berkay Ulker, Serap Ulusoy, Ertekin Utku Unal, Andrés Reyes Valdivia, Vincent Varley, Stewart Redmond

Collaborating Organisations: Vascular Society of Great Britain and Ireland (VSGBI), Research and Development Department at UHCW (sponsoring organisation), British Society of Endovascular Therapists (BSET), The Rouleaux Club (UK vascular trainee group), SingVasc (Singapore Vascular Surgical Collaborative), Vascupedia (European vascular education platform), Australian and New Zealand Vascular Trials Network (ANZVTN), Audible Bleeding (evidence based podcast, USA), British Society of Interventional Radiology (BSIR), the BSIR trainees (BSIRT), the European Society of Vascular Surgery (ESVS), the European Society of Vascular Trainees (ESVT), the Brazilian Society of Vascular Surgery (SBACV).

Reviewer acknowledgement: *JVSGBI* thanks the Editorial team for their contribution to the peer review of this work.

References

- Vascular and Endovascular Research Network (VERN) COVER Study Collaborative. Global impact of the first coronavirus disease 2019 (COVID-19) pandemic wave on vascular services. *Br J Surg* 2020;**107**(11):1396–400. https://doi.org/10.1002/bjs.11961
- Benson RA, Nandhra S. Outcomes of vascular and endovascular interventions performed during the coronavirus disease 2019 (COVID-19) pandemic: the Vascular and Endovascular Research Network (VERN) Covid-19 vascular service (COVER) tier 2 study. *Ann Surg* 2021;**273**(4):630–5. https://doi.org/10.1097/SLA.000000000004722
- Imray C. COVID-19 virus and vascular surgery, The Vascular Society for Great Britian & Ireland, 2020 [cited 2022]. Available from: https://www.vascularsociety.org.uk/professionals/news/113/covid19_virus_and_vascular_surgery.
- The Vascular and Endovascular Research Network (VERN) Committee. The COvid-19 Vascular sERvice (COVER) Study: an international Vascular and Endovascular Research Network (VERN) Collaborative Study assessing the provision, practice, and outcomes of vascular surgery during the COVID-19 pandemic. *Eur J Vasc Endovasc Surg* 2020;**60**(1):156–7. https://doi.org/10.1016/j.ejvs.2020.04.039
- Correia M, Constâncio V, Silva JC, *et al.* Is there an impact of COVID-19 on admission of patients to the emergency department for vascular surgery? *Ann Vasc Surg* 2020;69:100–4. https://doi.org/10.1016/j.avsg.2020.08.002
- Mouawad NJ, Woo K, Malgor RD, et al. The impact of the COVID-19 pandemic on vascular surgery practice in the United States. J Vasc Surg 2021;73(3): 772–9e4. https://doi.org/10.1016/j.jvs.2020.08.036
- Jacomelli J, Summers L, Stevenson A, Lees T, Earnshaw JJ. Update on the prevention of death from ruptured abdominal aortic aneurysm. *J Med Screening* 2017;**24**(3):166–8. https://doi.org/10.1177/0969141316667409
- Parkinson F, Ferguson S, Lewis P, Williams IM, Twine CP, Network SEWV. Rupture rates of untreated large abdominal aortic aneurysms in patients unfit for elective repair. *J Vasc Surg* 2015;61(6):1606–12. https://doi.org/10.1016/j.jvs.2014.10.023
- Conte MS, Bradbury AW, Kolh P, et al. Global vascular guidelines on the management of chronic limb-threatening ischemia. *Eur J Vasc Endovasc Surg* 2019;58(1):S1–109.e33. https://doi.org/10.1016/j.ejvs.2019.05.006
- Mills Sr JL, Conte MS, Armstrong DG, et al. The Society for Vascular Surgery Lower Extremity Threatened Limb Classification System: risk stratification based on wound, ischemia, and foot infection (Wlfl). J Vasc Surg 2014; 59(1):220–34.e1-2. https://doi.org/10.1016/j.jvs.2013.08.003

Appendix 1 Presenting features for each key condition

Aortic conditions

Aortic presentation	N (% of 194)	Diameter (cm) at presentation (mean (S.D))
AAA <5.5cm	31 (16.0%)	4.1 (1.2)
AAA ≥5.5cm (unruptured)	66 (34.0%)	7.4 (1.8)
Symptomatic / ruptured AAA	50 (25.8%)	7.6 (2.4)
Acute aortic syndrome and TAAA	47 (24.2%)	4.8 (1.7)

Acute limb ischaemia

Rutherford grade ⁽¹⁰⁾	N (% of 224)
1 - Viable	58 (25.9%)
2a - Salvageable if promptly revascularized	70 (31.2%)
2b - Salvageable if immediately revascularized	68 (30.3%)
3 - Non-salvageable	28 (12.5%)

Asymptomatic carotid

Presentation or indication	N (% of 33)
Incidental stenosis	18 (54.5%)
Bruit	4 (12.1%)
Pulsatile tinnitus	1 (3.0%)
Other	10 (30.3%)

Symptomatic carotid

Presentation	N (%)	<50%	50-69%	70-89%	90-99%	Near	Occlude
(modified		Stenosi	Stenosi	Stenosi	Stenosi	occiusio	a
Rankin ⁽¹¹⁾)		S	S	S	S	n	
Major	25	1	0	17	7	0	0
disabling							
Major non-	22	1	4	8	6	0	3
disabling							
Minor	16	0	2	4	10	0	0
Crescendo	34	0	8	14	11	1	0
TIA							
Single TIA	19	0	5	11	3	0	0
Monocular	0	0	0	0	0	0	0
TIA /							
amaurosis							
fugax							
Other	13	1	0	5	2	0	5

Chronic limb threatening ischaemia

Rutherford grade ⁽¹⁰⁾ :	N (% of 516)
Asymptomatic	20 (3.9%)
Mild claudication	27 (5.2%)
Moderate claudication	32 (6.2%))
Severe claudication	42 (8.1%)
Ischaemic rest pain	101 (19.5%)
Minor tissue loss	225 (43.5%)
Major tissue loss	69 (13.4%)

Diabetic foot infection

Wlfl ⁽¹²⁾ by category		Count (n)
Wound		
	0 Ischaemic rest pain / no ulcer or gangrene	3
	1 Minimal tissue loss	3
	2 Moderate tissue loss	93
	3 Extensive tissue loss	58
	Unknown	8
Ischaemia		
	0 No ischaemia	62
	1 Mild ischemia	29
	2 Moderate ischemia	34
	3 Severe ischaemia	36
	Unknown	72
Foot infection		
	0 No infection	0
	1 Superficial infections <2cm of cellulitis	9
	2 Moderate infection	113
	3 Severe infection, local with SIRS	44
	Unknown	3